Author's response to reviews

Title: A cross sectional study of upper extremity strength ten days after a stroke; relationship between patient-reported and objective measures

Authors:

Hanna C Persson (hanna.c.persson@vgregion.se)  
Anna Danielsson (anna.danielsson@neuro.gu.se)  
Katharina S Sunnerhagen (ks.sunnerhagen@neuro.gu.se)

Version: 13 Date: 28 August 2015

Author's response to reviews: see over
Gothenburg, 26th of August 2015

Dr James Burke
Associate Editor
BMC Neurology

Dear Dr Burke

Thank you for the comments on the manuscript, “A cross sectional study of upper extremity strength ten days after a stroke; relationship between patient-reported and objective measures”, MS 5303823527285372.

We have responded to the comments; please see the point by point response below. Changes are also marked in yellow within the manuscript.

1. Please place the List of abbreviations after the Conclusions section

   A list of abbreviations used in the manuscript is added after the conclusions.

2. It is not entirely clear from your manuscript whether this is the primary report of the SALGOT study or whether your primary findings will be published in a subsequent article. We would be grateful if you could clarify this in your cover letter.

   This study is a primary report from the SALGOT study. The SALGOT-study has more than one purpose, and will therefore have more than one study including primary findings. To clarify this in the manuscript, one sentence has been added in the beginning of the method.

Looking forward to hear from you soon,

Yours sincerely

Hanna C Persson

Corresponding author:

Hanna C Persson, PT, PhD-student,
Department of Clinical Neuroscience and Rehabilitation,
Institute of Neuroscience and Physiology,
Sahlgrenska Academy, University of Gothenburg, Sweden,
Telephone number: +46313423267,
E-mail: hanna.persson@neuro.gu.se